Author's response to reviews

Title: Adherence to Antiretroviral Therapy among HIV infected Children Measured by Caretaker Report, Medication Return, and Drug Level in Dar Es Salaam, Tanzania

Authors:

Frida W Mghamba (fwmghamba@gmail.com)
Omary MS Minzi (Minziobejayesu@gmail.com)
Augustine Massawe (draugustine.massawe@gmail.com)
Philip Sasi (Philip.sasi@gmail.com)

Version: 3 Date: 1 June 2013

Author's response to reviews:

MUHIMBILI UNIVERSITY OF HEALTH AND ALLIED SCIENCES
SCHOOL OF MEDICINE DEPARTMENT OF PAEDIATRICS AND CHILD HEALTH
P.O. BOX 65001 # DAR ES SALAAM # TANZANIA

Date: June 1, 2013

The Editor

Bio Medical Central Paediatrics Journal

Re: Resubmission of our manuscript for review and publication in your journal

Dear Editor BMC Paediatrics Journal, We thank you for efficient review of our manuscript with title Adherence to Antiretroviral Therapy among HIV infected Children Measured by Drug Level, Medication Returned and Caretaker Report in Dar-Es-Salaam, Tanzania. We have revised the manuscript after reading the reference 22, we agree with reviewers observation the frequency of CYP 2B6 detected in that study could influence the level of nevirapine in some patients in our study. The changes indicated in page 13 bottom and page 14 top.

Page 10, level of adherence. We have combined 2nd and 3rd sentence as you suggested.

We have review all the references and changes made in page 15-17. Example reference 22 previous version was reference 21 and Journal name Clinical Pharmacology Therapy 2010, 88(5):676-84
Kindly receive and handle our manuscript after addressing comments from reviewers in line with your guidelines. We confirm that the manuscript is an original work and there is no conflict of interest.

With kind Regards

Dr Mghamba F

Corresponding Author